Author's response to reviews

Title: Coffin Siris Syndrome with Mayer-Rokitansky-Kuster-Hauser syndrome

Authors:

Deepak Goyal (docdeepakgoyal@yahoo.com)
Dinesh K Yadav (dineshkumar169@yahoo.co.in)
Umesh Shukla (drumeshshukla@gmail.com)
Sidharth K Sethi (sidsdoc@gmail.com)

Version: 2 Date: 8 December 2009

Author's response to reviews: see over
To
The Editor
Journal of Medical Case Report

Sir

We are sending a case report titled “Coffin Siris Syndrome with Mayer-Rokitansky-Kuster-Hauser syndrome”. This association has probably never been described in medical literature. Kindly consider it for publication in your esteemed journal.

Regards

Dr Deepak Goyal
Registrar
Department of Pediatrics
PGIMER and associated Dr RML Hospital
Baba Kharag Singh Marg; Connaught Place
New Delhi
Ph: +91-9910580500
E mail: docdeepakgoyal@yahoo.com